Author's response to reviews

Title: Determinants of Intermittent preventive treatment of Malaria during pregnancy [IPTp] in a rural town in Western Nigeria

Authors:

Olorunfemi E Amoran Dr (drfamoran@yahoo.com)
Adebayo A Ariba Dr (adebayoariba@yahoo.com)

Version: 3 Date: 14 June 2012

Author's response to reviews:

highlighted text or tracked changes in word the amendments to the manuscript is hereby re-uploaded.